Author's response to reviews

Title: Characteristics of control group participants who increased their physical activity in a cluster-randomized lifestyle intervention trial

Authors:

Lauren A Waters (l.waters1@uq.edu.au)
Marina M Reeves (m.reeves@sph.uq.edu.au)
Brianna S Fjeldsoe (b.fjeldsoe@sph.uq.edu.au)
Elizabeth G Eakin (e.eakin@sph.uq.edu.au)

Version: 6 Date: 9 January 2011

Author's response to reviews:

Thankyou for allowing me to make four small typographical changes to the manuscript. The changes I have made are detailed below and highlighted in the manuscript.

(1) On page 13, "Usual care condition" has been changed to "Usual care group".

(2) On page 22 (reference list), text in reference 20 ("...A randomized controlled trial.") has been changed to bold.

(3) On page 23 (reference list), the journal volume and issue details have been added to reference 26 (", 29 (6): 566-573")

(4) On page 27 (table 2), "Income" has been changed to "Weekly household income".